Author’s response to reviews

Title: A case of misalignment: The perspectives of local and national decision-makers on the implementation of psychological treatment by telephone in the Improving Access to Psychological Therapy service

Authors:

Kelly Rushton (kelly.rushton@manchester.ac.uk)
Claire Fraser (claire.fraser@manchester.ac.uk)
Judith Gellatly (judith.l.gellatly@manchester.ac.uk)
Helen Brooks (helen.brooks@manchester.ac.uk)
Peter Bower (peter.bower@manchester.ac.uk)
Chris Armitage (chris.armitage@manchester.ac.uk)
Cintia Faija (cintia.faija@manchester.ac.uk)
Charlotte Welsh (charlotte.welsh@manchester.ac.uk)
Penny Bee (penny.e.bee@manchester.ac.uk)

Version: 2 Date: 06 Dec 2019

Author’s response to reviews:

Dr Kelly Rushton
School of Health Sciences
University of Manchester
Manchester, UK

Dr Christian Brettschneider
Associate Editor
BMC Health Services Research
Dear Dr Brettschneider,

RE: BHSR-D-19-01909 ‘A case of misalignment: The perspectives of local and national decision makers on the implementation of psychological treatment by telephone in the Improving Access to Psychological Therapy service’

Thank you for your email dated 5th December 2019, we are delighted the reviewers are happy with our revisions. We have now revised the manuscript in line with your editorial recommendations, which we hope will mean our manuscript might now be accepted for publication.

Revisions we have made to the Declarations section include:

1. The addition of an extra sentence regarding funders involvement in our study:
2. The addition of a statement confirming all authors have read and approved the final manuscript.
3. Re-ordering or the declarations section in line with the submission guidelines.

I hope these revisions meet with your approval and I would like to thank you for your time in reviewing our manuscript. Should you have any further queries, please do not hesitate to contact me.

Yours sincerely,

Kelly Rushton